Author’s response to reviews


Authors:

Etienne Audureau (Etienne.audureau@yahoo.fr)
James G Kahn (jgkahn@ucsf.edu)
Marie-Hélène Besson (Marie-Helene.Besson@axiosint.com)
Joseph Saba (Joseph.saba@axiosint.com)
Joël Ladner (Joel.Ladner@chu-rouen.fr)

Version: 2 Date: 28 May 2012

Author’s response to reviews:

In Methods, one sentence has been added regarding data authorization, as follows: "All data analyses were conducted under the authorization from Boehringer Ingelheim."